Author’s response to reviews

Title: Randomized Pilot Trial of a Synbiotic Dietary Supplement in Chronic HIV-1 Infection

Authors:

Marco Schunter (Marco.Schunter@rbk.de)
Hiutung Chu (hiuchu@caltech.edu)
Timothy L Hayes (tlhayes@emory.edu)
Delandy McConnell (ayanamiyoo@gmail.com)
Sean S Crawford (cartassc@gmail.com)
Paul A Luciw (paluciw@ucdavis.edu)
Stig Bengmark (stig@bengmark.se)
David M Asmuth (david.asmuth@ucdmc.ucdavis.edu)
Jennifer Brown (jennifer.brown@ucdmc.ucdavis.edu)
Charles L Bevins (clbevins@ucdavis.edu)
Barbara L Shacklett (blshacklett@ucdavis.edu)
J William Critchfield (jwcritchfield@ucdavis.edu)

Version: 2 Date: 14 March 2012

Author’s response to reviews: see over
March 14, 2012

BMC Complementary and Alternative Medicine
BioMed Central
The Open Access Publisher

Re: Revised Manuscript Submission

Dear BMC Complementary and Alternative Medicine,

Attached please find our manuscript entitled Randomized Pilot Trial of a Synbiotic Dietary Supplement in Chronic HIV-1 Infection. It has been revised in response to the editorial requests you provided us following initial submission on March 7, 2012.

Specifically, we have 1) Included a CONSORT checklist, 2) Added a Competing Interests section after the conclusions, 3) Added an Author Contributions section following Competing Interests, and 4) Checked to make sure that the Abstract is found on both the manuscript document and on the submission system, and conforms to the formatting guidelines.

We hope that these changes enable you to initiate the review process. If there are issues remaining, please let us know and they will be addressed promptly. Once again, thank you for considering our manuscript.

With appreciation and on behalf of all authors,

Bill

J. William Critchfield, Ph.D.
Associate Project Scientist
Medical Microbiology and Immunology
School of Medicine
University of California, Davis
1 Shields Avenue
Davis, CA 95616 USA
Tel: +1 (530) 754-9611
jwcritchfield@ucdavis.edu